Author's response to reviews

Title: Identification of the first intragenic deletion of PITX2 gene causing an Axenfeld-Rieger-Syndrome: Case report

Authors:

Guillaume de la Houssaye (delaho@necker.fr)
Ivan Bieche (ivan.bieche@pharmacie.univ-paris5.fr)
Olivier Roche (roche@necker.fr)
Veronique Vieira (vieira@necker.fr)
Ingrid Laurendeau (ingrid.laurendeau@pharmacie.univ-paris5.fr)
Laurence Arbogast (arbogast@necker.fr)
Hatem Zeghidi (Zeghidi@necker.fr)
Philippe Rapp (Rapp@necker.fr)
Philippe Halimi (halimi@aphp.fr)
Michel Vidaud (michel.vidaud@pharmacie.univ-paris5.fr)
Maurice Menasche (menasche@necker.fr)
Jean-Louis Dufier (dufier@necker.fr)
Marc Abitbol (abitbol@necker.fr)

Version: 5 Date: 30 August 2006

Author's response to reviews:

Dear EDITOR in CHIEF,

With the Help of Guillaume de La HOUSAYE we have increased the resolution of the Figure 4 and thus we resubmit the manuscript slightly modified without changing the legend of the Figure 4. We leave it as it was initially

Warm thanks for understanding What we tried in order to meet as soon as possible the requirements of the reviewers

Very Sincerely

Marc Abitbol MD.PhD